Author's response to reviews

Title: Burden of frailty in the elderly population: perspectives for a public health challenge

Authors:

Fanny BUCKINX (fanny.buckinx@ulg.ac.be)
Yves ROLLAND (rolland.y@chu-toulouse.fr)
Jean-Yves REGINSTER (jyreginster@ulg.ac.be)
Celine RICOUR (celine.ricour@chu.ulg.ac.be)
Jean PETERMANS (Jean.petermans@chu.ulg.ac.be)
Olivier BRUYERE (olivier.bruyere@ulg.ac.be)

Version: 3
Date: 4 February 2015

Author's response to reviews: see over
Editor's comment
Title: Burden of frailty in the elderly population: perspectives for a public health challenge
Authors:
Fanny BUCKINX (fanny.buckinx@ulg.ac.be)
Yves ROLLAND (rolland.y@chu-toulouse.fr)
Jean-Yves REGINSTER (jyreginster@ulg.ac.be)
Celine RICOUR (celine.ricour@chu.ulg.ac.be)
Jean PETERMANS (Jean.petermans@chu.ulg.ac.be)
Olivier BRUYERE (olivier.bruyere@ulg.ac.be)
Version: 2 Date: 3 February 2015

Editor’s comment:
Dear authors,
we thank you for your reply to the reviewers and the way you incorporated the changes in the manuscript.
Although your manuscript is not a systematic review, you started your literature search by the selection of close to 2800 articles. Therefore I think that your manuscript could improve and benefit enormously if you could include a Prisma figure (see prisma guidelines for systematic reviews).

Author’s response:
We fully agree with the editor. The flowchart of the literature search (according to prisma guidelines) was added to the manuscript.